Author's response to reviews

Title: How protective is cervical cancer screening against cervical cancer mortality in developing countries? The Colombian case.

Authors:

Luz Angela Choconta-Piraquive (lachocontap@bt.unal.edu.co)
Nelson Alvis-Guzmán (nalvis@yahoo.com.mx)
Fernando De la Hoz-Restrepo (fpdelahozr@unal.edu.co)

Version: 2 Date: 15 December 2009

Author's response to reviews:


Dear editorial board:

On behalf of my co authors I am submitting the revised manuscript for you to consider its publication in BMC Health Services Research.

We shortened the abstract, background, results and discussion while focusing on table 1 and the most important findings of our analysis.

According to your recommendations we tried to improve the methods section and make them clearer.

We declared this article has not been submitted for evaluation or published in any other journal. All authors listed have read the final manuscript and agreed to its submission.